Author's response to reviews

Title: Age and gender specific normal values of left ventricular mass, volume and function for gradient echo magnetic resonance imaging: a cross sectional study

Authors:

- Peter A Cain (peteracain@gmail.com)
- Ragnhild Ahl (ragnhild.ahl@skane.se)
- Erik Hedstrom (erik.hedstrom@med.lu.se)
- Martin Ugander (martin.ugander@med.lu.se)
- Ase Allansdotter-Johnsson (ase.johnsson@vgregion.se)
- Peter Friberg (peter.friberg@mednet.gu.se)
- Hakan Arheden (hakan.arheden@med.lu.se)

Version: 4 Date: 9 August 2008

Author's response to reviews:

Dear Editor,

We have identified a minor error in the calculation of one of the results. The only changes that have been made to the attached manuscript are numeric changes which can be found on page 9 and are highlighted in green. The changes do not affect the conclusions.

Regards,

Martin Ugander, submitting author